Author's response to reviews

Title: Evaluation of the PaO2/FiO2 ratio after cardiac surgery as predictor of outcome during hospital stay.

Authors:

Francisco Esteve (f.esteve@bellvitgehospital.cat)
Juan C Lopez-Delgado (juancarloslopezde@hotmail.com)
Casimiro Javierre (cjavierre@ub.edu)
Konstantina Skaltsa (kskaltsa@ub.edu)
Maria LL Carrio (lcarrio@bellvitgehospital.cat)
David Rodriguez-Castro (drodriguez@bellvitgehospital.cat)
Herminia Torrado (htorrado@bellvitgehospital.cat)
Elisabet Farrero (elfarrero@bellvitgehospital.cat)
Josep LL Ventura (6775jvf@comb.cat)
Antonio Diaz-Prieto (adiazprieth@bellvitgehospital.cat)
Rafael Mañez (rmanez@bellvitgehospital.cat)

Version: 2  
Date: 26 April 2014

Author's response to reviews: see over
To the Editorial Team of *BMC Anesthesiology,*

On behalf of my co-authors, I submit the enclosed manuscript for consideration by the Journal. It has not been published in this or a substantially similar form (in print or electronically, including on a web site), nor accepted for publication elsewhere, nor is it under consideration by another publication. There are not possible conflicts of interest (including financial and other relationships) for each author.

Appropriate Ethics Committee approval has been obtained for the research reported (if you need committee writing documents we would be glad to show yours).

All authors send this letter as confirmation that they have read and approved the paper, have met the criteria for authorship as established by the International Committee of Medical Journals Editors, believe that the paper represents honest work, and are able to verify the validity of the results reported.

I hope it will be of your interest. Thank you for this opportunity of submitting our report.

If you need further information, please contact us,

Sincerely, on behalf of all authors,

Juan Carlos López Delgado  
**Hospital Universitari de Bellvitge**  
**Servei de Medicina Intensiva**  
C/Feixa Llarga s/n.08907.  
L’Hospitalet de Llobregat.  
juancarloslopezde@hotmail.com